Author's response to reviews

Title: Advance lung cancer inflammation index (ALI) at diagnosis is a prognostic marker in patients with metastatic non-small cell lung cancer (NSCLC): A retrospective review

Authors:

Syed H Jafri (sjafri@lsuhsc.edu)
Runhua Shi (RShi@lsuhsc.edu)
Glenn M Mills (GMills@lsuhsc.edu)

Version: 2 Date: 8 February 2013

Author's response to reviews: see over